Author's response to reviews

Title: Community-based infant hearing screening in a developing country: parental uptake of follow-up services

Authors:

Bolajoko O Olusanya (boolusanya@aol.com)
Oladele O Akinyemi (dele_akinyemi@excite.com)

Version: 6 Date: 17 February 2009

Author's response to reviews: see over
17 February 2009

The Editor  
BMC Public Health 
BioMed Central  
London, UK

Dear Sir,

Revised Manuscript MS: 9229650211913650  
Community-based infant hearing screening in a developing country: parental uptake of follow-up services

In the process of reviewing the proof, we found it necessary to carry out some minor editing for better clarity.

No addition information or data has been added as you will notice from the tracked changes.

The colours will be converted to black when the indicated changes are accepted.

Yours sincerely

Bolajoko O. Olusanya MBBS, PhD, FRCPCH